Author’s response to reviews

Title: The diagnostic accuracy of anti-CCP antibody in rheumatoid arthritis: Impact of reporting quality on pooled diagnostic estimates

Authors:

Elias Zintzaras (zintza@med.uth.gr)
Afroditi Papathanasiou (apapath@med.uth.gr)
Dimitrios Ziogas (dziogas@bidmc.harvard.edu)
Michael Voulgarelis (mvoulgar@med.uoa.gr)

Version: 4 Date: 13 May 2012

Author's response to reviews: see over
Dear Editor,

We were pleased to hear that BMC Musculoskeletal Disorders has accepted our manuscript for publication subject to a minor administrative revision. All comments have been addressed in this revised version. In particular, the following readings were added after the Discussion section:

“Competing interests
The authors declare that they have no competing interests. “

“Authors’ contributions
EZ conceived and supervised the study; EZ and AP drafted the manuscript; AP extracted the data and performed the statistical analysis; DZ and MV extracted the data and performed the statistical analysis of the revised manuscript; EZ and MV drafted the revised manuscript.”

Acknowledgements
Scientific support for this project was provided through the Biometrical, Epidemiological and Clinical Research Organization (BECRO).

We thank you again for the opportunity to improve our manuscript. Having made this revision, we hope that the manuscript would be acceptable for publication in BMC Musculoskeletal Disorders in its current version.

Sincerely yours,

Elias Zintzaras